Author’s response to reviews

Title: A Novel Germline ARMC5 Mutation in a Patient with Bilateral Macronodular Adrenal Hyperplasia: a case report

Authors:
Qiuli Liu (liuqiuli900827@163.com)
Dali Tong (tongdali1985@163.com)
Jing Xu (jingxu8023@163.com)
Xingxia Yang (yang_xing_xia@163.com)
Yuting Yi (yiyt@geneplus.org.cn)
Dianzheng Zhang (dianzhengzh@pcom.edu)
Luofu Wang (wangluofu@aliyun.com)
Jun Zhang (zhjunkk@163.com)
Yao Zhang (zy921-1@163.com)
Yaoming Li (finleylym@126.com)
Lianpeng Chang (changlp@geneplus.org.cn)
Rongrong Chen (chenrr@geneplus.org.cn)
Yanfang Guan (guanyf@geneplus.org.cn)
Xin Yi (yix@geneplus.org.cn)
Jun Jiang (jiangjun_64@163.com)

Version: 1 Date: 15 Jan 2018

Author’s response to reviews:

Dear editor,

We have revised the paper as you have suggested. I confirmed that all author details on the revised version are correct, that all authors have agreed to authorship and order of authorship for
this manuscript and that all authors have the appropriate permissions and rights to the reported data.

Thank you for your consideration.

Sincerely,

Jun Jiang, MD, PhD

Department of Urology

Institute of Surgery Research

Daping Hospital, Third Military Medical University

Chongqing 400042, PR China